Author's response to reviews

Title: Feasibility Outcomes of a Presurgical Randomized Controlled Trial Exploring the Impact of Caloric Restriction and Increased Physical Activity versus a Wait-List Control on Tumor Characteristics and Circulating Biomarkers in Men Electing Prostatectomy for Prostate Cancer

Authors:

Wendy Demark-Wahnefried (demark@uab.edu)
Jeffrey W Nix (jnix@uabmc.edu)
Gary R Hunter (ghunter@uab.edu)
Soroush Rais-Bahrami (sraisbahrami@uabmc.edu)
Renee A Desmond (rdesmond@uabmc.edu)
Balu Chacko (bchacko@uab.edu)
Casey D Morrow (caseym@uab.edu)
Maria Azrad (maria123@uab.edu)
Andrew D Fruge (fruge@uab.edu)
Yuko Tsuruta (tsuru@uab.edu)
Travis Ptacek (tptacek@uab.edu)
Scott A Tully (stully@urological.com)
Roanne Segal (RSegal@Ottawahospital.on.ca)
William E Grizzle (wgrizzle@uab.edu)

Version: 3
Date: 12 December 2015

Author's response to reviews:

See attached cover letter that addresses the three minor revisions requested.